Author's response to reviews

Title: An unusual presentation of multiple cavitated lung metastases from colon carcinoma.

Authors:

Patrizia Seminara (patrizia.seminara@uniroma1.it)
Gaia Manna (gaia.manna@hotmail.it)
Alessandra Emiliani (emiliani.alessandra@libero.it)
Alessandro Iannace (ale_iann@hotmail.com)
Tania Losanno (tania.los@libero.it)

Version: 2 Date: 8 April 2011

Author's response to reviews:

We have revisited the article following the reviewer's comment:

1. Pag.2: More clinical information of case (Tumor serum markers and inflammatory indices (erythrocytes sedimentation rate, C-reactive protein) were normal. White blood cells count (9,680 cells/mcL) was within the normal reference range.)

2. The use of impersonal form in Abstract, Background, and Discussion

3. Correction of one reference (reference 5: Gaeta M et al.)

4. The English form